Author's response to reviews

Title: Ethical Issues In Implementation Research: A Discussion of The Problems In Achieving Informed Consent

Authors:

Jane L Hutton (J.L.Hutton@warwick.ac.uk)
Martin P Eccles (martin.eccles@ncl.ac.uk)
Jeremy M Grimshaw (jgrimshaw@ohri.ca)

Version: 3 Date: 5 December 2008

Author's response to reviews:

Dear Editor,

We have accepted all of the copy editor's suggestions throughout the article and have clarified the two phrases he identified as being unclear.

Dr Wensing raised two discretionary revisions that we were asked to respond to. The first was the structure of the paper and the second was whether the title accurately reflected the article content.

In the response to the point about the structure of the paper while we don't disagree we are following the specified article structure for a debate article from the instructions to authors. Under the circumstances we have chosen to make any further changes.

In the response to the comment about the title we have changed the title to "Ethical Issues In Implementation Research: A Discussion of The Problems In Achieving Informed Consent" in order to better represent the discussive nature of the article.

We hope that with the changes the article is now acceptable for publication.

Yours Faithfully,

Martin Eccles